Author's response to reviews

Title: Delay to celiac disease diagnosis and its implications for health-related quality of life

Authors:

Fredrik Norström (fredrik.norstrom@epiph.umu.se)
Lars Lindholm (lars.lindholm@epiph.umu.se)
Olof Sandström (olof.sandstrom@vll.se)
Katrina Nordyke (katrina.nordyke@epiph.umu.se)
Anneli Ivarsson (anneli.ivarsson@epiph.umu.se)

Version: 4 Date: 3 November 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production team,

We were informed to make the following changes of our paper:

“Formatting changes to be made:
-----------------------------
Major revisions (we require the author to make these changes)

Figures: Figure legend 2 contains graphics. These will either need to be included within the image as a key, or written in text (e.g. "black square"). Please ensure the corrected legend does not exceed 250 words.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Additional Material: please rename this section 'Additional Files' and place it after the figure legends.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.”

As a consequence of the suggested changes have we included the image part of Figure legend 2 as legend of Figure 2.

We have also moved the section ‘Additional Material’ and changed title to ‘Additional Files’ as requested.

In our revision we have also decreased the font of the legend in Figure 1.

Yours sincerely,

Fredrik Norström

on behalf of all co-authors